Author's response to reviews

Title: Consistency between stated and revealed preferences: a Discrete Choice Experiment and a behavioral experiment on vaccination behavior compared.

Authors:

Mattij S Lambooij (mattijs.lambooij@rivm.nl)
Irene A Harmsen (irene.harmsen@rivm.nl)
Jorien Veldwijk (jorien.veldwijk@rivm.nl)
Hester de Melker (hester.de.melker@rivm.nl)
Liesbeth Mollema (liesbeth.mollema@rivm.nl)
Yolanda WM van Weert (Yolanda.van.weert@rivm.nl)
Ardine AG de Wit (ardine.de.wit@rivm.nl)

Version: 2
Date: 5 December 2013

Author's response to reviews: see over
Dear professor Pala,

Hereby I resubmit the paper ‘Consistency between stated and revealed preferences: a Discrete Choice Experiment and a behavioral experiment on vaccination behavior compared.’ For publication in BMC Medical Research Methodology.

This revision is a reaction to a request by Ms Pura, asking to send appendix A. Apparently, I had forgotten this appendix during the initial submission. I apologize for this incompleteness of the submission.

This version is identical to the previous submission, I only added appendix A to the files.

Sincerely yours,

Mattijs Lambooij, PhD.
Dept. Quality of Care and Health Economics
National Institute of Public Health and the Environment
PO box 1
3720 BA, Bilthoven
The Netherlands
Mattijslambooij@rivm.nl